PRImary care Streptococcal Management (PRISM) study: in vitro study, diagnostic cohorts and a pragmatic adaptive randomised controlled trial with nested qualitative study and cost-effectiveness study

Paul Little,^{1*} FD Richard Hobbs,² Michael Moore,¹ David Mant,³ Ian Williamson,¹ Cliodna McNulty,⁴ Gemma Lasseter,⁴ MY Edith Cheng,¹ Geraldine Leydon,¹ Lisa McDermott,¹ David Turner,¹ Rafael Pinedo-Villanueva,¹ James Raftery,¹ Paul Glasziou³ and Mark Mullee¹ on behalf of the PRISM investigators

¹Primary Care and Population Sciences Unit, University of Southampton, Southampton, UK ²Primary Care Clinical Sciences, University of Birmingham, Birmingham, UK ³Department of Primary Care Health Sciences, University of Oxford, Oxford, UK ⁴Public Health England, Primary Care Unit, Microbiology Laboratory, Gloucestershire Royal Hospital, Gloucester, UK

*Corresponding author

Declared competing interests of authors: Paul Little is a member of the NIHR Journals Library Board. James Raftery is a member of the HTA Editorial Board and NIHR Journals Library Editorial Group.

Published January 2014 DOI: 10.3310/hta18060

Plain English summary

PRImary care Streptococcal Management (PRISM) study Health Technology Assessment 2014; Vol. 18: No. 6 DOI: 10.3310/hta18060

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

General background

The overuse of antibiotics in general practices, mostly for illnesses such as sore throat, chest infections and ear infections, is potentially a big problem for us all for several reasons. First, antibiotic overuse increases the risk of antibiotic resistance, whereby bacteria become resistant to antibiotics and are no longer killed by antibiotics. This could potentially lead to serious infections as a result of 'superbugs' becoming untreatable both now and for future generations. Antibiotics commonly cause side effects such as allergic reactions, diarrhoea and skin rashes. Using them also increases people's belief in them – because they think it is the antibiotics that helped them get better, when in fact they would have got better in the same time anyway. This leads people to think that they need to come back the next time they get an infection – so it 'medicalises' illnesses, uses NHS resources and also exposes patients to unnecessary antibiotics.

Background – the context for sore throats

Antibiotics are still prescribed for most patients with a sore throat attending their general practitioner (GP) or nurse in primary care. This is despite the best available evidence, which suggests there is a modest benefit overall from antibiotics. One approach to tackle this is to target antibiotics better, using a simple 'clinical score' – whereby doctors or nurses prescribe according to particular symptoms and examination findings. Another approach is to use rapid antigen detection tests (RADTs), which are very commonly used in many countries. To use a RADT, a swab is taken from the throat, and the RADT gives a quick answer as to whether the most important bacteria are present or not. The particular type of bacteria that RADTs pick up is a common type of streptococcus bacteria – called Lancefield group A haemolytic streptococcus (GABHS). This bacterium can cause both a sore throat and more serious illnesses.

However, there are problems with using either a clinical score or a RADT:

- There is debate about which RADT should be used and how.
- It is unclear whether other bacteria (other than GABHS) are important, particularly streptococci from other groups Lancefield groups C and G. RADTs will not pick up these other bacteria.
- For a clinical score, it is not clear which symptoms and examination findings most clearly tell us whether bacteria are present.
- There have also been very few good studies that compare RADTs with clinical scores, or with other approaches, such as delayed antibiotic prescribing. Delayed prescribing is where the patient is advised to use an antibiotic after several days if symptoms are not starting to settle.

The PRImary care Streptococcal Management (PRISM) study was made up of several substudies that tackled these issues:

Laboratory study

If rapid antigen tests are to be used for patients in everyday practice, they have to be accurate, easy to use, inexpensive and potentially widely available. Several such tests are available, and in the first study five RADTs were tested in the laboratory with different types and concentrations of bacteria. One of the best of these was the IMI test, which was both one of the most accurate and found to be relatively easy to use.

[©] Queen's Printer and Controller of HMSO 2014. This work was produced by Little *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Clinical study – developing a clinical score

Two large groups of patients (606 in the first group, 517 in the second) came to see the doctor or nurse with a sore throat and agreed to take part. Their symptoms and signs were documented and a throat swab was sent to the laboratory to see if bacteria were present. The results showed that patients who had Lancefield groups C or G bacteria had the same kind of illness as those with group A strains. It was also possible to develop a useful clinical score to help pick up the main types of bacteria (A, C or G) based on a simple count of five items. The five items make up the acronym FeverPAIN:

- Fever during the last 24 hours
- Pus (white spots) on the tonsils
- coming quickly to see the doctor within 3 days (Attend rapidly)
- very Inflamed tonsils
- and **N**o cough or runny nose.

Trial of clinical scores and rapid antigen detection tests

The trial compared three ways of managing sore throat among 1760 patients who came to see their doctor:

- 1. Delayed antibiotic prescribing group (the control group).
- 2. Clinical score group: the score was worked out and antibiotics were advised for high scores. No antibiotics were advised for low scores, and delayed antibiotics for those in the middle. The first clinical score that was developed (score 1; n = 1129) was replaced by a more valid score (FeverPAIN; n = 631) as the trial went on.
- 3. RADT group: the clinical score was also worked out. For low and middle scores, the plan was similar to that used in the clinical score group. A RADT was used for those with high scores, and, if the result was positive, antibiotics were advised and, if the result was negative, no antibiotics were given.

The study found that using the clinical score (FeverPAIN) improved control of symptoms, and both the clinical score and the RADT reduced antibiotic use. Moderately bad or worse symptoms resolved significantly faster (30% faster) in the clinical score group but not in the RADT group (11% faster).

Health economic analysis

If RADTs were to be used more widely, it would be important to show that using them is a cost-effective use of time and money for the health service. The study showed that using RADTs was probably more expensive and less cost-effective than using the clinical score.

Qualitative study

Face-to-face and telephone interviews were done with 51 people – GPs, nurse practitioners and patients from general practices across Hampshire, Oxfordshire and the West Midlands. Patients and nurses were very positive about using clinical scores and RADTs. Doctors had a number of concerns about both RADTs and clinical scores that would need to be addressed before widespread implementation would work – particularly related to the perceived usefulness of clinical scores in the face of clinical experience and intuition.

Conclusions

There are RADTs that are not expensive, easy to use and are potentially widely available for use in primary care. Although they will detect GABHS, RADTs are not designed to detect other strains such as Lancefield C or G strains. Lancefield C or G strains commonly cause streptococcal sore throats, and patients have a similar illness to those who have A strains. A five-item score (acronym FeverPAIN) to predict streptococcal infection is likely to be valid but further validation is preferable. When antibiotics are targeted using a clinical score (FeverPAIN), this improves control of symptoms, reduces antibiotic use and is very cost-effective. Using a RADT in addition to using the clinical score provides no clear benefits for patients over using the clinical score alone. RADT use is also more costly, probably less cost-effective and faces several barriers from clinicians. To implement the use of clinical scores more widely in everyday practice will require addressing the issues doctors have.

© Queen's Printer and Controller of HMSO 2014. This work was produced by Little *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Five-year impact factor: 5.804

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, The Cochrane Library and the ISI Science Citation Index and is assessed for inclusion in the Database of Abstracts of Reviews of Effects.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: nihredit@southampton.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

The HTA programme, part of the National Institute for Health Research (NIHR), was set up in 1993. It produces high-quality research information on the effectiveness, costs and broader impact of health technologies for those who use, manage and provide care in the NHS. 'Health technologies' are broadly defined as all interventions used to promote health, prevent and treat disease, and improve rehabilitation and long-term care.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

For more information about the HTA programme please visit the website: www.hta.ac.uk/

This report

The research reported in this issue of the journal was funded by the HTA programme as project number 05/10/01. The contractual start date was in October 2006. The draft report began editorial review in August 2012 and was accepted for publication in January 2013. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health.

© Queen's Printer and Controller of HMSO 2014. This work was produced by Little *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

Editor-in-Chief of *Health Technology Assessment* and NIHR Journals Library

Professor Tom Walley Director, NIHR Evaluation, Trials and Studies and Director of the HTA Programme, UK

NIHR Journals Library Editors

Professor Ken Stein Chair of HTA Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

Professor Andree Le May Chair of NIHR Journals Library Editorial Group (EME, HS&DR, PGfAR, PHR journals)

Dr Martin Ashton-Key Consultant in Public Health Medicine/Consultant Advisor, NETSCC, UK

Professor Matthias Beck Chair in Public Sector Management and Subject Leader (Management Group), Queen's University Management School, Queen's University Belfast, UK

Professor Aileen Clarke Professor of Health Sciences, Warwick Medical School, University of Warwick, UK

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Peter Davidson Director of NETSCC, HTA, UK

Ms Tara Lamont Scientific Advisor, NETSCC, UK

Professor Elaine McColl Director, Newcastle Clinical Trials Unit, Institute of Health and Society, Newcastle University, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Honorary Professor, Business School, Winchester University and Medical School, University of Warwick, UK

Professor Jane Norman Professor of Maternal and Fetal Health, University of Edinburgh, UK

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professorial Research Associate, University College London, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Please visit the website for a list of members of the NIHR Journals Library Board: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: nihredit@southampton.ac.uk